Author's response to reviews

Title: Nutrition and lung cancer: a case control study in Iran

Authors:

Mostafa Hosseini (mhossein110@yahoo.com)
Parisa Adimi Naghan (faster20@gmail.com)
Ali Moghadas Jafari (alimoghaddas7@yahoo.com)
Mahmoud Yousefifard (yousefifard20@gmail.com)
Shervin Taslimi (shervin_883@yahoo.com)
Kian Khodadad (yousefifard@razi.tums.ac.ir)
Forouzan Mohammadi (mahmood_ano@yahoo.com)
Makan Sadr (faster200@gmail.com)
Mansour Rezaei (yousefifardm@gmail.com)
Mohammad Reza Masjedi (mrmasjedi@nritld.ac.ir)

Version: 5
Date: 10 October 2014

Author's response to reviews: see over
Dear respected Editor

Many thanks for your kindly comment and initial acceptance of our manuscript. Below point-by-point we answered the invaluable matters you commented. We hope that these answers are enough for publishing our paper.

Best,
Prof. Mostafa Hosseini
Department of Epidemiology and Biostatistics,
Faculty of Public Health and Institute of Public Health Research,
Tehran University of Medical Sciences, Tehran, Iran

You wrote: Nutrients are not simply food but the ingredients of the food. The authors are advised at least state in the Discussion section what the compounds are or may be that are related to the causation or prevention of lung cancer in Iran.

Answer: Many thanks. To address this, several bits are added to Discussion section and highlighted as “red”.

In the Conclusion part (including the Abstract), the authors have to add "in Iran" for the food items they mentioned to distinguish the raw materials and products from other regions of the world.

Answer: Many thanks. Done.